RESEARCH ARTICLE

Conversion of asymptomatic infection to symptomatic visceral leishmaniasis: A study of possible immunological markers

Vidya Nand Rabi Das[®]*[®], Sanjiva Bimal[®], Niyamat Ali Siddiqui, Ashish Kumar, Krishna Pandey, Sanjay Kumar Sinha, Roshan Kamal Topno, Vijay Mahentesh, Ashish Kumar Singh, Chandra Shekhar Lal, Subhankar Kumar Singh, Pradeep Das

Rajendra Memorial Research Institute of Medical Science (ICMR), Patna, India

• These authors contributed equally to this work.

* drvnrdas@yahoo.com

Abstract

Introduction

Presence of asymptomatic individuals in endemic areas is common. The possible biomarkers in asymptomatic individuals once they get exposed to infection as well as following conversion to symptomatic disease are yet to be identified. We identified asymptomatic Visceral leishmaniasis (VL) infection amongst rK39⁺sorted direct agglutination test positive (DAT⁺) endemic healthy population and confirmed it by quantitative PCR(qPCR). The immunological determinants such as Adenosine deaminase (ADA), Interferon gamma (IFN- γ), Tumour Necrosis Factor alpha (TNF- α) and Interleukin 10 (IL-10)were examined to predict probable biomarkers for conversion to symptomatic VL.

Methods

Sample size was 5794 healthy individuals from VL endemic region. Antibody tests(DAT &rK39) were performed and later a qPCR assay was employed using kDNA specific primers and probes. Immunological biomarkers examined were ADA level by ADA–MTP kit and quantitative cytokines(IFN- γ , IL-10 and TNF- α) by ELISA.

Results

120 asymptomatic individuals of 308 rK39 sero-positives were DAT positive comprising of 56 with previous history and 64 with no history of VL. RT-PCR confirmed asymptomatic VL in 42 sero-positives. These were followed up through repeated qPCR and evaluation of immunological determinants. We observed10 symptomatic cases converted from a total of 42 asymptomatic individuals identified at base-line. The level of ADA, IL-10 and IFN- γ remained consistently high in asymptomatic cases and amongst these, ADA and IL-10 but not IFN- γ remained higher at the development of clinical symptoms into active VL. On the contrary, there was no significant change in the mean concentration of TNF- α at both stages of the disease.



GOPEN ACCESS

Citation: Das VNR, Bimal S, Siddiqui NA, Kumar A, Pandey K, Sinha SK, et al. (2020) Conversion of asymptomatic infection to symptomatic visceral leishmaniasis: A study of possible immunological markers. PLoS Negl Trop Dis 14(6): e0008272. https://doi.org/10.1371/journal.pntd.0008272

Editor: Claudia Ida Brodskyn, Centro de Pesquisa Gonçalo Moniz-FIOCRUZ/BA, BRAZIL

Received: September 10, 2019

Accepted: April 7, 2020

Published: June 18, 2020

Copyright: © 2020 Das et al. This is an open access article distributed under the terms of the <u>Creative</u> <u>Commons Attribution License</u>, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All relevant data are within the manuscript and its Supporting Information files.

Funding: The author(s) received no specific funding for this work.

Competing interests: No authors have competing interest.

Discussion

We surmise from our data that considerable proportion of asymptomatic cases can be a reservoir and may play a crucial role in transmission of visceral leishmaniasis in endemic areas.

The data also suggests that ADA and IL-10 can serve as a potential biomarker during the conversion of asymptomatic into symptomatic VL.

Author summary

The most threatening form of leishmaniasis is human visceral leishmaniasis, which is caused by L. donovani in Indian subcontinent. The disease accounts for huge annual burden of infectious disease in India. Efforts towards disease elimination programme are far beyond satisfaction and there is need to re-strengthen health monitoring, surveillance programme in endemic areas. There is a greater need to identify asymptomatic individuals amongst endemic healthy control by active surveillance programme. Further, except a few, many asymptomatic subjects become sero-negative without developing VL due to strong inherent cellular immunity possessed by them. This study enumerates use of rK39, DAT and quantitative PCR(qPCR) in asymptomatic detection and then evaluate immunological biomarkers (Adenosine deaminase, IFN- γ , TNF- α and IL-10) which help in disease conversion to symptomatic VL, at 6 month follow- up. Follow-up analysis revealed that antibody testing with rK39 (3.24%), DAT (8.33%) had poor agreement with disease conversion. By qPCR, we could identify the level of parasitemia, and collectively with serology, the test detected 23.80% of asymptomatic cases converted into disease. The level of ADA and IL-10 remained consistently high during conversion. IFN-γ became high in asymptomatic infection but became low on conversion, whereas, no significant change in level of TNF- α was observed at both stages of disease. Thus by use of qPCR, concomitantly with DAT over rK39⁺ sorted endemic healthy control may be employed for early case detection. Adenosine is produced by ecto-nucleotide during *L.donovani*infection, has been previously reported and in parallel to this study we conclude that determination of adenosine deaminase level (ADA) might help identify early cases with more tendency to convert into disease. This will help to boost health monitoring programme to eradicate Kala-azar in Bihar, India.

Introduction

Leishmaniasis is a disease caused by protozoan parasite of genus *Leishmania* caused by 20 *Leishmania* species which are transmitted to humans by the bites of infected *Phlebotomus* sand flies. In Indian sub-continent, Kala-azar or human visceral leishmaniasis (VL) caused by *L. donovani* is most fatal if left untreated. In India, the disease is compounded further by toxicity, drug resistance, relapse and appearance of post kala-azar dermal leishmaniasis (PKDL) in 5–15 years after treatment [1,2]. A major proportion of many exposed persons, harbour the parasite but remain the asymptomatic leaving only a small number of individuals to exhibit clinical manifestation [3, 4]. Therefore asymptomatic cases are wide spread in VL endemic areas [5,6]. Asymptomatic cases were also earlier reported to harbour the parasite in their blood [7].

In the absence of animal reservoir, asymptomatic VL and PKDL patients in particular, are thought to be reservoir for visceral leishmaniasis in Indian sub-continent and this necessitates an accurate diagnosis of such cases prevalent in endemic areas [8, 9]. Previously, the direct demonstration of parasite was based on microscopy or culture of blood and aspiration(bone marrow, lymph-nodes or spleen). The sensitivity of such tests varies considerably and is associated with risks of serious bleeding especially the splenic aspiration and should only be carried out at settings with surgical services. Although many specific diagnostic tests for confirmed VL patients exist, but these tests do not address the time and proportion of asymptomatic individuals who will be clinically manifest [3, 5]. This necessitates comprehensive longitudinal studies to identify the components which will dictate the conversion from asymptomatic to symptomatic VL. The diagnosis of VL has been further advanced by molecular tools such as PCR [10] and qPCR (<10% to up to 30%) [11, 12]. More importantly, the most of asymptomatic individuals are identified from areas, where VL remains endemic and even exceed the number of symptomatic cases [13-18]. Such cases assume importance as probable reservoir of VL transmission in endemic areas but on account of ethical issues and invasiveness, these tests cannot be identify the time of conversion of these subjects into symptomatic and the associated components which can influence conversion[3,5]. Many immunological methods such as direct agglutination test (DAT) and lateral flow immunochromatographic tests, such as rK39 and rkE-16 have been introduced to screen large number of individuals in endemic areas [19–23]. Commercially available rapid diagnostic tests are sensitive and specific with acceptable reproducibility and most are heat stable (WHO/ TDR, 2011). Such serological tests accompanied with a PCR assay can be well applied in identification of asymptomatic individuals living in endemic areas.

The issue, how only a few exposed asymptomatic individuals but not the all develop full blown disease manifestations is still unexplored. One study showed the identical pattern of immune response of asymptomatic cases and cured VL but there were certain changes in the level of ADA in asymptomatic individuals versus cured VL [24, 25].

The Main thrust of the study was to identify the key immunological Biomarkers for the emergence of asymptomatic cases and the factors that lead to their conversion into clinically manifestation of the disease. The assay of ADA activity in the serum and other biological fluids is reported to be useful in many pathological conditions [26, 27]. Cell mediated immunity (CMI) plays an important role, where causative organism grows intra-cellularly for instance, VL caused by *L. donovani* and reports also have suggested importance of ADA as a potential marker of cellular immunity.

Mononuclear cells are known to secrete ADA in, response to *Leishmania* infection [28] and our previous reports have also shown that marginal numbers of asymptomatic individuals develop cell mediated immunity and the elevated serum activity for ADA is associated in conversion of asymptomatic cases into symptomatic VL [1]. Immuno-pathogenesis of VL is determined by cytokines produced during cellular immune response to identify the subsequent resistance or susceptibility of a VL patient to disease [29]. The inability of a VL patient to mount protective Th¹ immune response (IL-2 IFN- γ , TNF- α), exacerbates the pathogenicity and in some cases results in resistance to drug [1, 30]. There are reports that when over expression of IL-10, render the resistant mice, susceptible to infection [31]. Both the IL-10 and IL-6 response is triggered which upregulates in all forms of leishmaniasis and their level subsides as the patients recover from the disease [32,33].

Attempt has been made in the present study to elucidate the key immunological factors that promote the conversion of asymptomatic individuals to symptomatic stage of visceral leishmaniasis. We demonstrate that a seizable proportion of asymptomatic cases can be reservoir and may play a crucial role in transmission of visceral leishmaniasis in endemic areas.

Materials and methods

Study site and sample size

Two endemic areas in Bihar state: Kothia and Pirari, Saran, two known villages with high endemicity for VL, were selected to conduct this study. The ethics committee of the Institute had approved the project and informed consent was taken from all subjects included in the study. Total population covered for initial screening was 9445 which were clinically examined for fever, present and past history of kala-azar, weight loss, splenomegaly and hepatomegaly.

Inclusion criteria

Participants of either sex aged more than or equal to (\geq) 6 years with or without family history of VL were included from the sampled villages.

Exclusion criteria

Exclusion criteria were haemoglobin(Hb) <5 g/100 mL; white blood count <1,000/mm³; thrombocyte count <50,000; alanine transaminase, aspirate transaminase, and alkaline phosphatase \geq 2 times upper limit of normal, serum creatinine or blood urea nitrogen \geq 1.5 times upper limit of normal. Patient'ssero-positive for HIV, hepatitis B and C, tuberculosis (TB), malaria, kidney disease, hypersensitivity to amphotericin B or inactive ingredients of the amphotericin B formulation were also excluded. The rK39-negative patients were classified as non-kala-azar cases and managed by the testing physician for an alternative illness.

Experimental design

Following preliminary clinical and haematological investigations all were screened by rK39, immuno-chromatographic test (ICT) and the asymptomatic cases who defined an endemic areas individual with rK39 positive without signs and symptoms were segregated under three-categories of anti-leishmanial antibody titre category 1 (<1:800), category 2 (1:1600–1:6400) and category 3 (>1:6400). These categories were based on DAT result.Among the subjects category 1 and category 2, DAT positive sero-positives underwent confirmation of VL by qPCR and immunological tests (ADA level, IFN- γ , IL-10 and TNF- α levels). The subjects were followed up clinically, parasitologically (qPCR) and immunologically (Adenosine deaminase Activity, IFN- γ , IL-10 and TNF- α) at 6- month follow up.

For all these investigations initially, one spot of blood from all selected individuals including the healthy controls the treated and untreated VL patients were immediately tested, using an rk-39 strip. Further, samples of figure prick blood from these subjects was sent at immunology division of Rajendra Memorial Research Institute of Medical Sciences (ICMR-RMRIMS), Patna for assessment of the DAT titre. After obtaining informed consent, 5ml whole blood was also collected by venepuncture in a plain and sodium-EDTA vacutainer (BD Biosciences, USA) from each study subject. Serum was extracted for both rK-39 and the DAT while plasma was extracted from sodium-EDTA vacutainer for quantitative assessment of cytokine through sandwich ELISA.

rK39 strip test

Kala-azar detect, a commercial version of the immuno-chromatographic strip test based on the rK39 antigen (In Bios International, Seattle, WA) was initially applied for screening all 5794 subjects. Briefly, the test was performed on a drop of blood, placed on the absorbent pad on nitrocellulose membrane, followed by addition of two drops of chase buffer. Results were considered positive when, two pink lines(one control and one test) appeared after 10 minutes.

DAT

The antigen for the DAT was prepared and test performed as previously reported [23, 34]. For this, around 5 mm punched filter with dried blood spot(whatman filter paper4) was kept overnight in DAT diluent for elution. A sample was considered positive if it had a titre \geq 1:800[4].

Determination of ADA level

Determined the ADA activity by ADA-MTP kit (Tulip Diagnostic India) according to the instructions illustrated in manufacturer protocol. In brief, fresh blood samples weretaken out in sodium- EDTA vacutainer (BD Bioscience) which wascentrifuged at 3000rpm for 2 min at 4°C. The upper plasma was then transferred to a sterile cryo vial, flash froze in liquid nitrogen and stored at -80°C till further use. The fluorescent intensity was determined at Ex/Bn 535/587 by carry Eclipse Fluorescance Spectrophotometer (Agilent technologies, CA, USA). The same plasma protein concentration was used in each assay.

ELISA analysis for different cytokine (IFN-γ, IL-10& TNF-α)

The total level of cytokines(IFN- γ , IL-10& TNF- α) was measured by BD OptiEIA (BD Bioscience SD, USA) ELISA kits using the culture supernatants of antigen stimulated PBMC culture and post setup of culture according to instructions provided by the manufacturer. The colour intensity was measured at 450nm by iMARK Microplate Reader (Biorad).Cells (PBMCs) were obtained following Histopaque 1077(Sigma Helrich, Dorset, UK) density gradient centrifugation at 400g for 30 min at 20°C from the heparinised peripheral blood of the subjects. Following washing twice in RPMI 1640, PBMCs were suspended in complete RPMI medium at a concentration of 1×10^6 cells/ml. These cells (1×10^6 cells/ml) were cultured at a concentration of 2×10^5 cells/ml in 24-well tissue culture plates in complete medium for 16h at 37°C and 5% Co₂. Cell culture supernatants were analysedfor cytokines by ELISA techniques according to the manufacturer's instruction [35]. The detection limits were 5.6pg/ml for TNF- α (EMD Milipore). All samples were simultaneously run in triplicate cells. Cytokine absorbence was read at a wavelength of 450nm in an enzyme-linked immunosorbent assay(ELISA) reader (Bio-Rad, iMark, Gurgaon, India).

Quantitation of parasites in asymptomatic patients(qPCR)

DNA was isolated from blood using a QIAamp DNA blood mini kit (Qiagen, Hilden, Germany) as per the manufacturer's instructions. Blood sample were kept in ATL buffer and protease K and incubated at 56°C for 2 hours before further processing according to the manufacturer's instructions. Extracted DNA samples were stored at -20°C until real-time PCR.To determine the parasitic load, we performed the Real Time PCR (Roche) using SYBR green (Roche) chemistry by (kDNA) using a kDNA gene-specific primer (F-5'-TCTGTGGCCCATT TGTTGTA-3',andR-5'-CATTTTTGGGTTTTCGGAGA-3').To quantify the parasitic load of each sample was extrapolated to standard curve using 10 fold serially diluted *L. donovani* parasites DNA corresponding to 10^4 to 0.1parasites per reaction as shown in our previous paper [36-39].The cycling conditions were as follows: 1 cycle at 95°C for 3 min and 40 cycles of 95°C for 15 s (denaturation), 58°C for 30 s (annealing), and 72°C for 30 s (extension). The fluorescence signal was captured at the end of each cycle using the SYBR channel (490-nm wavelength for excitation and 525-nm wavelength for emission).The threshold cycle value (Ct) was calculated for each sample by determining the point at which fluorescence exceed the thresholds limit and calculated the parasites/µg DNA.

Data management, quality assurance and statistical analysis

The entire questionnaire was scrutinized for accuracy and consistency. Frequency and distribution checks were applied to access the range of values and identify missing data. Checks were also applied during designing the Data Entry Program to reduce the data inaccuracy. All forms of suspected cases were double-entered and errors validated against original record. In addition, RMRI field team was looked into the entire data collection process using independent vehicles, and was conducted systematic back checks. Finally, the medical teams visit clinically and haematologically suspected subjects and reconfirmed that they do indeed meet the definition of suspected cases. All the structured/semi-structured questionnaires used under this survey were pre-tested. After verification, each record was given to data entry operators for data entry. Data were entered, cleaned using Epi Info Version 3.3.2 (CDC Atlanta).

Data was extracted from the structured/semi-structured questionnaires using a structured data sheet specially designed for the purpose. Data extraction was conducted by the project workers, working for the project. Before embarking on the data collection process, field workers attended a two-days training provided by the principal investigator on how to fill the structured/semi-structured questionnaires. To ensure data quality, the following measures were taken: (a) additional two-days training was given to field supervisor before the start of data extraction, (b) the overall activities of data extraction was monitored by the principal investigator, and there was strict supervision during data collection also, (c) all completed datasets were examined by the principal investigator for completeness, and (d) from the data extracted from each structured/semi-structured questionnaires, 5% of the sample was randomly selected and validated against the registers/records by the principal investigator. The extracted data was verified from two sources i.e. from door-to-door visit of clinically and haematologically suspected subjects.

Data were analyzed using Graph Pad Prism 5 (USA software) and presented as Mean \pm SD. Student's two-tailed paired t-test or unpaired t test was used to determine the difference between the groups studied. P value <0.05 were considered statistically significant for all the analysis.

Results

Baseline field survey

Based on health monitoring government data, it was assumed that about 10,000 population was required to be surveyed to get sizeable number of asymptomatic cases from highly VL affected district of Bihar, India. As such, one highly endemic village was identified within the identified PHC i.e.Garkha and Dariapur. House to house survey was conducted to collect demographic data through pretested questionnaires. Proper informed consent was obtained from head / representative of the family.

Altogether 5794 individuals who were aged ≥ 6 years in the selected villages were screened with rk-39 strip in order to identify the baseline asymptomatic carriers of *L. donovani* infection. Out of that, 2835 (49%) was male and 2959 (51%) was females. Among them, maximum (38%) belonged to 6–15 years of age, followed by 25% in the age group of 16–30 years. The minimum and maximum age of the participant was 6 and 72 years respectively with mean age of 37 years. A total of 308 individuals were found to be rk-39 sero-positive which comprised 188 individuals having past history of kala-azar. Remaining 120 sera samples from a total of 308 rk-39 positive individuals had no previous history of kala-azar.

Detection of sub-clinical infection with L. donovani

Based on the anti-leishmania antibody titres observed in the DAT, those 308 rk-39 sero-positive subjects were divided into three groups; those that were DAT negatives (<1:800) (DAT neg), a positive DAT with a low titre range (1: 1600–1: 6400) (DAT posLow) and a positive DAT $(>1:6400; DAT_{pos}High)$ with maximum titre range. Based on such categorization, we identified 24 asymptomatic suspects under the DAT Pos Low / titre range, 30 asymptomatic subjects under the DAT PosHigh titre range and 10 suspects under DAT Neg titre range. Fifty six (56) individuals, showing rk-39 positivity out of 308 suspects were having previous history of kala-azar, comprising (N = 30) DAT $_{Pos}$ Low titre range and (N = 18) DAT $_{pos}$ High titre range. Comparison of parasite load byqPCR with 56 individuals with previous history of Kala-azar revealed that out of 56 such individuals, 15 in DAT_{Pos}Low and 8 in DAT_{Pos} High were qPCRpositive.We found direct correlation for increased DAT titre range with Ld/µg DNA in qPCR. For instance, where rK-39 positive suspects were in DAT $_{Pos}$ Lowtitre, the average $Ld/\mu g$ DNA was >94 whereas, when it increased (DAT>1:6400), the average Ld DNA/ μ g became \geq 232 Ld/ μ g DNA. Similarly, in individuals with no previous history of VL, out of 64 subjects, we observed 6 qPCR positive individuals in DAT PosLowand 13 qPCR positive in DAT PosHigh. Based on such categorization, we could finally identify 42 asymptomatic suspects (Table 1, Fig 1).

Monitoring of serum ADA level in VL patients, endemic asymptomatic subjects and in endemic asymptomatic subjects converted into VL at 6 months follow-up

In a supplementary experiment it was shown that serum ADA activity in VL patients at before treatment remained significantly high compared to after treated and healthy subjects (Table 2, Fig 2 p<0.0001).

Serum ADA activity in endemic healthy individuals was observed low and in normal range. It has also previously been shown that ADA is secreted against pathogens [28] and it also observed elevated during VL [1], thereby, we anticipated that fluctuated and altered ADA activity should possibly reflect the presence of parasite in a few asymptomatic cases, being converted into VL. As such, we monitored serum ADA activity in endemic symptomatic subjects before and after conversion (Table 3, Fig 2). As discussed in the preceding section, we had identified a total of 10 rK-39 suspects who had shown high anti-Leishmania DAT titre, were also qPCR positive. Assuming those 10 suspects as asymptomatic cases of VL from endemic area of VL in Bihar state, India, we next tried to know the status of ADA activity in such endemic asymptomatic subjects (Table 3, Fig 2). The data revealed that all 10 qPCR positive asymptomatic subjects were found to possess elevated serumADA level (Table 3, Fig 2 p = 0.0284).

Serum ADA level in VL patients in their pre and post treatment stage and endemic asymptomatic subjects

VL patient group produced significantly higher level of serum ADA at pre-treatment stage than was observed to the corresponding value of ADA in their respective post treatment stage and healthy control (Table 2, p<0.0001). Anticipating that some of the asymptomatic subjects might lead to clinical manifestation of VL with increasing time period, we examined serum ADA activity of asymptomatic individuals. Results obtained were interesting and ADA activity observed was not consistent, between the initial survey and after 6-month follow-up, ten (10) subjects (All of whom were subsequently found p-PCR positive and sero-positives) developed VL manifestation such as irregular fever, malaise, loss of weight and splenomegaly. The RMRIMS's clinicianshadparasitologically confirmed these individuals asVL. Sera samples of

Subjects	Subjects rK39 positive suspects (120)			Asymptomatic to VL	
	DAT	Number	qPCR negative (Ld/µg	qPCR Positive (Ld/	conversion
	Anti- Leishmania titre (Baseline)		DNA)	µgDNA)	(6 months)
Past history of VL	<1:800	8	-	-	-
(n = 56)	1:1600-1:6400	30	15	15	3
·	>1:6400	18	10	8	4
No history of VL $(n = 64)$	<1:800	10	-	-	-
	1:1600-1:6400	24	18	6	
	>1:6400	30	17	13	3
Confirmed VL (n = 50) Control	<1:800	-	-	-	NA
	1:1600-1:6400	30	-	30	
	>1:6400	20	-	20	

Table 1. Development of endemic healthy control into symptomatic VL at the end of 6 month follow up (rK39, DAT and qPCR analysis).

 \bullet <40, Ld/µg DNA was considered negative samples.

• Only qPCR positive individuals were considered for further studies.

https://doi.org/10.1371/journal.pntd.0008272.t001

such cases was found to produce very high ADA level to their respective baseline ADA activity (p = 0.0284). The remaining 32 asymptomatic subjects who were found qPCR positive at baseline did not complain of fever or any other symptom during the study period.

Cytokine trend in asymptomatic VL and during conversion from asymptomatic to symptomatic VL with appearance of clinical manifestations of visceral leishmaniasis

As illustrated in Fig 3, control group produced relatively low IL-10, IFN- γ and TNF- α . Endemic asymptomatic patients produced significantly higher quantitative yield of IL-10,



https://doi.org/10.1371/journal.pntd.0008272.g001

Subject	Number	ADA activity(U/L)	P v	alue
Healthy control	10	46.54±9.07	<0.0001	
VL untreated	28	222.10±54.29		<0.0001
VL treated	23	61.77±13.23		

Table 2. Serum ADA activity of active	VL cases before and after treatment (Serum A	DA activity (A) of healthy subject (n = 10), VL untreated $(n = 28)$ and VL
treated $(n = 23)$).			

https://doi.org/10.1371/journal.pntd.0008272.t002

while IL-10 produced by non-converted asymptomatic suspects (DAT, rK39 positive and RT-PCR positive n = 32) was significantly low. As soon as, such asymptomatic patients developed full clinical manifestations of VL(6 months), IL-10 produced by these patients once again increased and became 1.36 fold higher compared to quantity of IL-10 produced by asymptomatic suspects, not converted into VL, and in them, IL-10 was over all 7.31 fold down.

On the other hand, both categories of asymptomatic VL (converted verses non-converted) individuals produced significantly higher IFN- γ then were observed in healthy subjects. Comparatively, the IFN- γ levels in asymptomatic VL patients were higher at base-line then in non-converted asymptomatic suspects. The scenario became reverse and unlike to IL-10, there was observed decrease of about 2.40 fold in total yield of IFN- γ , when converted patients follow-up at 6 months on appearance of full blown clinical manifestations. The mean level of IFN- γ in non-converted endemic asymptomatic individuals was at base-line and at 6 month follow-up remained low and comparable.



Fig 2. Serum ADA activity of healthy subject (n = 10), VL untreated (n = 28), VL treated (n = 23), Non-converted (n = 32) asymptomatic VL and after conversion into VL (n = 10) (Before conversion and after conversion (6 month)). Non-significant (ns), ***P < 0.001, *P < 0.05.

https://doi.org/10.1371/journal.pntd.0008272.g002

Asymptomatic VL (rK3	θ +, DAT+, qPCR+) (n = 42)	ADA activity(U/L)	P value	
Non-conv	erted (n = 32)	148.90 ± 42.47	0.9642	
Converted into VL (n = 10)	Before conversion	149.70 ± 75.26		0.0284
	After conversion (6 months)	242.30 ± 116.40		

Table 3. Serum ADA activity in asymptomatic VL and after conversion into VL (Serum ADA activity of asymptomatic VL (n = 10) and after conversion into active VL (n = 10)).

• P-value 0.9642 corresponds to before conversion set of data.

https://doi.org/10.1371/journal.pntd.0008272.t003

Asymptomatic VL patients consistently were observed to produce higher yield of TNF- α before and after conversion. On the other hand, asymptomatic individuals who failed to convert into clinical VL were observed with low yield of TNF- α which however was statistically non-significant.

Discussion

We identified potential biomarkers in VL patients, whose levels may be correlated with the conversion of asymptomatic patients to become symptomatic up to clinical manifestation and disease. Advancement towards the disease is likely due to increased *L.donovani*, which may modulate host CMI response. Many VL patients become immuno-compromised and die due to secondary infection. After the treatment failure somehow facilitates the problem as early in



Fig 3. The relationship between cytokine production (IL-10, IFN- γ and TNF- α) and positivity in RT-PCR indicating subclinical infection and during the development of symptomatic visceral leishmaniasis (VL) in 10 asymptomatic residents of an endemic area of Bihar. Healthy subject (n = 10), asymptomatic VL (non-converted (n = 32) and converted into VL (n = 10)).Data were represented as mean ± SD, P \leq 0.05 were considered as significant.* P \leq 0.05, **P \leq 0.01, ***P \leq 0.001, ns (non-significant).

https://doi.org/10.1371/journal.pntd.0008272.g003

asymptomatic cases not identified. We assessed the pattern of cellure immune response in the community in VL endemic areas to predict the presence of sub-clinical infection [5]. Cyto-kines that regulate cellular immunity are the critical checkpoints to determine resistance or susceptibility to disease [40, 41]. Amongst cytokines IFN- γ has a pivoted role in protection in all forms of leishmaniasis [29, 42]. This cytokine works in tandem with TNF- α to exert leishmanicidal activitythrough reactive oxygen species (ROS) and reactive nitrogen species (RNS) generation are available [8, 43]. VL patients, however, release inadequate IFN- γ and develop various clinical conditions such as splenomegaly pancytopenia, anemia and disseminated hemorrhages [8, 31, 44–46] in the presence of excess IL-10 production [25, 32, 47, 48]. Adenosine produced by macrophages has a critical role in IL-10 production during *L.donovani* infection [49]. Adenosine was reported to suppress the release of IL-12 and nitric oxide production during infection [28,50]. Therefore, we measured serum ADA activity and cytokine profile (IL-10, TNF- α , and IFN- γ) at baseline and at 6-month follow-up in a community.

Direct diagnostic approaches for parasite detection in either spleen or bone marrow smears are usually not preferred for community screening due to invasiveness and ethnic consideration, hence diagnosis gets delayed. Antibody detection kits are usually used (rK39, rk26) for sero- surveillance [4, 23]. We identified 308 individuals out of 5794 screened population as rK39 strip test positive. Specifically, the rK39 strip test is commonly used worldwide, despite its low sensitivity in Africa [51]. More over sensitivity of rK39 in Sudan is only 67% [52]. Therefore, we also further validated such sero-positives through direct agglutination test (DAT). We performed DAT as the next step detection method for several reasons. This test is an easy serological method and even less skilled health workers can perform it at primary health care level [19, 53, 54]. More importantly, the inclusion of 2- Mercaptoethanol in the diluents makes DAT more compatible to identify L.donovani specific IgM antibodies which get expressed at the earliest, soon after individuals get exposed to the parasite [55, 56]. Our concern in this study was straight forward to detect and filter asymptomatic individuals with VL from a large number of the population residing in the VL endemic area. Before applying DAT for community screening, we had performed a Chequer board analysis, where an anti-leishmania titer of 1:800 was observed optimal as the cut-off with 91.7% sensitivity and 100% specificity for specific diagnosis of VL [4]. DAT analysis gave a validated proof that only 120 individuals out of 308 rK39sero-positives were exposed in the real sense, in whom DAT become able to identify parasite-specific antibodies. The few false positives recorded by rK39 amongst the community can be due to advanced or very early infection. While the strip test appears less specific than the DAT, it is recommended to use the combined application of rK39 and DAT for optimizing identification of very early (asymptomatic) infection of VL. Whether or not, these 120 sero-positives (rK39, DAT), actually represented asymptomatic VL?. This question arises due to certain drawbacks of antibody-based diagnostic tools which sometimes become positive in recovered VL subjects at different time span after cure.

In subsequent experiments, we went on to perform quantitative PCR on serologically filtered 120 DAT positive asymptomatic VL subjects. In similar to our findings, some previous studies also suggested that transmission of *L. donovani* in Bihar is focal (120 of the 308 rK39 positive, who lived in the endemic region of Bihar); 56 lived in households with a family history of VL and 64 lived in households with no history of VL. Anyway, both groups (those with history of VL and those with no history of VL), living in the same endemic region, were generally at high risk of infection.

Forty-two (42) seropositive suspects out of 102 DAT positive showed qPCR positivity and all had recorded more than 40 Ld/mg/DNA. From qPCR analysis, unlike sero-positives, we get evidence for the first time in this study that suspected individuals living in households with family history, were at particularly high risk of infection. The usefulness of quantitative PCR in

epidemiology has recently been shown for the early detection of visceral leishmaniasis cases [11, 12, 57, 58]. This analysis helped us to identify 42 asymptomatic VL subjects and it necessitated to explore, how many of them converted into disease as well as the factors that promoted conversion of asymptomatic individuals to VL cases. The follow-up analysis revealed that most (32 of the 42 qPCR positives) cleared their infection within 6 months. Many asymptomatic subjects become sero-negative, without developing VL, within 12–36 months [5]. It has been shown earlier that some of them showing serological reactivity to *Leishmania* possess strong cellular immunity, and this helps to clear parasite from their blood before inducing disease in such cases [59]. Curiously, among the 42 qPCR positive subjects also found sero-positive at the baseline survey, only 7(seven) of the 23 from households with a family history of VL and 3 of the 19 from households with no family history of VL developed the symptomatic disease at 6 months.

Therefore altogether 10 of 42 (23.80%) asymptomatic cases having rk-39, DAT and qPCR positivity converted into disease during this study project. This approach appears to be more authentic because if the rk-39 strip test and DAT would be considered, individually as sole criteria for baseline survey, chances of detecting difficult cases with parasitemia in resource-poor rural areas become low. As a result, only 3.27% (10 of 308) and 8.33% (10 of 120), considered positive at baseline converted from asymptomatic to symptomatic VL. The study period remained during the period of VL transmission (February- November) and it is anticipated that most *L.donovani* infections, which convert to disease, take a time of 6 months in the endemic areas of Bihar [4]. We surmise from our data that qPCR done on samples, filtered initially based on rk-39 and DAT can be useful in the detection of early parasitemia and conversion to symptomatic VL.

In our study those 10 patients who converted into VL among them the minimum Ct value is ~32. Asides host, parasites were also modulating the host cellular immune response during conversion from early parasitemia in asymptomatic VL patients to established parasitemia in grown-up VL patients with various clinical manifestations. Several reports have also suggested that parasitic micro-organisms seek the help of adenosine and adenosine receptors during the internalization of parasite and to skip the protective immune response of the host [60, 61]. It would be interesting to add here that there are two types of ADA activities that operate under the influence of iso-enzymes in the host during an inflammatory response. One iso-enzyme, ADA-1, is responsible for protective immune response and is present in all tissues. In contrast to it, we reported the presence of A2B receptors in monocytes and macrophages on VL patients and the majority of such A2B+ expressing monocytes were in general high IL-10 producers [62]. The course of ADA activities observed in VL patients is due to the activation of other principal iso-enzymes, ADA-2. ADA-2 is only present on monocytes and macrophages and its activation is triggered in response to the pathogen [28, 63].

The serum adenosine deaminase (ADA) activity was therefore assessed as this can potentially influence host suppressive Th2 immune response (IL-10) during the progression of early parasitemia in asymptomatic individuals. Initial testing was done at RMRIMS, Patna in VL patients before and after treatment. The findings reflected more ADA activity in active and untreated VL patients compared to in VL patients after treatment and control. Asymptomatic patients identified at baseline screening were observed with consistently high serum ADA activity. The rest of the individuals had their ADA level to be par with the normal range as observed in control. ADA activities remained high in both categories of DAT and rK39 positive subjects are they were qPCR positive or negative. Therefore even at the onset of the immune response, when individuals get exposed demonstrate threshold antibodies to *Leishmania* antigen after biting of infected sandfly and in them, ADA level gets increased. Earlier also, it has been reported that serum ADA activities can be used as a marker to monitor the progression of asymptomatic to symptomatic VL.

We reported earlier the necessity of A2B adenosine receptors in the severity of VL infection [62]. Evidence was also drawn during this study that when monocytes were infected in the presence of adenosine, the rate of *L. donovani* infection increased, whereas *Leishmania* infection became abrogated after blocking the adenosine receptor [62]. Our result corroborates the previous finding in context to the precise link between load and ADA levels. More interestingly, the production of IL-10 remained significantly high in the group of asymptomatic patients, both before and after conversion than in the group of asymptomatic patients, who did not convert into VL. These findings seem to be very interesting when ADA activities in such patients are also considered.

Previous reports suggest that adenosine produced by ectonucleotidase, during activation of the ATP pathway, during infection contributes to the production of IL-10 and IL-6 [62].

We reported earlier the crucial role between IL-6 and up-regulated surface 3-ectonucleotidase activities synergizing with the anti inflammatory effect of IL-10 [48, 64] and STAT 3 phosphorylation [65]. Considering the growth and differentiation of inducing properties, IL-6 might also appear to correlate with increased ectonucleotidase activities in converted asymptomatic patients. We, however, did not examine the IL-6 level and concentrated on IL-10 but can hypothesize that IL-6 promotes the anti- inflammatory effect of IL-10 in leishmaniasis [51].The level of IFN- γ remained high, until the patients converted into symptomatic VL, and showed significantly low production. There was no change in the level of TNF- α during conversion into symptomatic patients.

Taken together above results suggest that combined application of rk-39, DAT and R T– PCR can help detect asymptomatic VL patients and also that asymptomatic VL patients have a big role in the VL transmission cycle in resource-poor rural areas of Bihar. Targeting adenosine and detecting ADA level can help us to monitor the progression of patients from early parasitemia in asymptomatic VL to clinical disease, since IL-10 was observed in high concentration, during conversion into clinical disease; IL-10 level concurrent to serum ADA activity can help to facilitate early intervention and interruption of disease transmission. Although the cohort tested in our study did not include population-specific genotype markers but the inclusion of such parameters could indeed foster prediction by combining molecular diagnostics with serodiagnostic. In one of the recent most studies on asymptomatic VL, were able to link several HLA-DRβ allele groups with the progression of VL [66].

Ethics statement

This study was approved by the Institutional Ethics Committee of RMRIMS, Patna vide approval No.01/IEC/2014 dated 28thJanuary, 2014.Participants of either sex aged more than or equal to (\geq) 6 years with or without family history of VL were included in this study. The study specific written informed consent was obtained from the study participant. In case of minor (>6 years and <18 years), written informed consent was obtained from their parents/ guardians.

Acknowledgments

We are thankful to the Chief Medical Officer of Saran, Medical Officer In-charge of concerned PHCs for extending their support and co-operation for smooth execution of this study. Technical services rendered by Mr. Santosh Kumar Sinha, Technician, Department of Immunology, Mr. Madan Mohan Azad, Data Manager, Clinical Medicine is highly acknowledged and appreciated.

Author Contributions

Conceptualization: Vidya Nand Rabi Das.

Data curation: Niyamat Ali Siddiqui, Sanjay Kumar Sinha, Ashish Kumar Singh.

Formal analysis: Sanjiva Bimal, Niyamat Ali Siddiqui, Sanjay Kumar Sinha, Ashish Kumar Singh.

Investigation: Sanjiva Bimal, Ashish Kumar, Roshan Kamal Topno, Vijay Mahentesh, Subhankar Kumar Singh.

Methodology: Niyamat Ali Siddiqui, Roshan Kamal Topno.

Project administration: Pradeep Das.

Supervision: Krishna Pandey, Sanjay Kumar Sinha, Pradeep Das.

Validation: Niyamat Ali Siddiqui, Krishna Pandey, Chandra Shekhar Lal.

Visualization: Vidya Nand Rabi Das.

Writing – original draft: Vidya Nand Rabi Das, Sanjiva Bimal, Chandra Shekhar Lal, Subhankar Kumar Singh.

Writing - review & editing: Krishna Pandey.

References

- 1. Vijayamahantesh, Amit A, Dikhit MR, Pandey RK, Singh K, Mishra R, Das VN, Das P, Bimal S(2016). Elevated Serum ADA Activity as a Marker for Diagnosis and Prognosis of Visceral Leishmaniasis and Post Kala-Azar Dermal Leishmaniasis in Indian Patients. PLoS One. 2016 May 17; 11(5):e0154117. eCollection 2016. <u>https://doi.org/10.1371/journal.pone.0154117</u> PMID: 27186641
- Zijlstra EE, Musa AM, Khalil EA, el-Hassan IM, el-Hassan AM (2003): Post-kala-azar dermal leishmaniasis. Lancet Infect Dis. 2003 Feb; 3(2):87–98. <u>https://doi.org/10.1016/s1473-3099(03)00517-6</u> PMID: <u>12560194</u>
- Singh S, Kumari V, Singh N (2002). Predicting kala-azar disease manifestations in asymptomatic patients with latent Leishmania donovani infection by detection of antibody against recombinant K39 antigen. Clin Diagn Lab Immunol. 2002 May; 9(3):568–72. <u>https://doi.org/10.1128/cdli.9.3.568-572.</u> 2002 PMID: <u>11986261</u>
- Bimal S, Singh SK, Das VN, Sinha PK, Gupta AK, Bhattacharya SK, Das P(2005). Leishmania donovani: effect of therapy on expression of CD2 antigen and secretion of macrophage migration inhibition factor by T-cells in patients with visceral leishmaniasis. Exp Parasitol. 2005 Oct; 111(2):130–2. <u>https:// doi.org/10.1016/j.exppara.2005.05.009</u> PMID: <u>16023642</u>
- Schaefer KU, Kurtzhals JA, Gachihi GS, Muller AS, Kager PA (1995). A prospective sero-epidemiological study of visceral leishmaniasis in Baringo District, Rift Valley Province, Kenya. Trans R Soc Trop Med Hyg. 1995 Sep-Oct; 89(5):471–5. https://doi.org/10.1016/0035-9203(95)90070-5 PMID: 8560511
- Sinha PK, Bimal S, Pandey K, Singh SK, Ranjan A, Kumar N, Lal CS, Barman SB, Verma RB, Jeyakumar A, Das P, Bhattacharya M, Sur D, Bhattacharya SK(2007). A community-based, comparative evaluation of direct agglutination and rK39 strip tests in the early detection of subclinical Leishmania donovani infection. Ann Trop Med Parasitol. 2008 Mar; 102(2):119–25. <u>https://doi.org/10.1179/136485908X252278 PMID: 18318933</u>
- Goswami RP, Bairagi B, Kundu PK (2003). rK39 strip test—easy, reliable and cost-effective field diagnosis for visceral leishmaniasis in India. J Assoc Physicians India. 2003 Aug; 51:759–61. PMID: <u>14651134</u>
- Singh Ashish Kumar, Rabi Das Vidya Nand, Amit Ajay, Dikshit Manas R, Mahentesh Vijaya, Kumar Akhilesh, Pandey Raj Kishore, Narayan Shyam, Singh Bipin K, Pandey Krishna, Verma Neena, Das Pradeep, and Bimal Sanjiva (2018). Identification of Clinical Immunological Determinants in Asymptomatic VL and Post Kala-azar Dermal Leishmaniasis Patients. Iran J Parasitol. 2018 Oct-Dec; 13(4): 541– 548. PMID: <u>30697307</u>
- 9. GANGULY Swagata, et al. PKDL—a silent parasite pool for transmission of leishmaniasis in kala-azar endemic areas of Malda District, West Bengal, India. PLoS neglected tropical diseases, 2015, 9.10.

- Schallig HD, Schoone GJ, Kroon CC, Hailu A, Chappuis F, Veeken H (2001). Development and application of 'simple' diagnostic tools for visceral leishmaniasis. Med Microbiol Immunol. 2001 Nov; 190(1– 2):69–71. https://doi.org/10.1007/s004300100083 PMID: 11770114
- Moulik S, Chaudhuri SJ, Sardar B, Ghosh M, Saha B, Das NK, Chatterjee M. Monitoring of parasite kinetics in Indian post–kala-azar dermal leishmaniasis. Clinical Infectious Diseases. 2017 Sep 13; 66 (3):404–10.
- Sengupta R, Chowdhury SJ, Moulik S, Ghosh MK, Saha B, Das NK, Chatterjee M. Active surveillance identified a neglected burden of macular cases of Post Kala-azar Dermal Leishmaniasis in West Bengal. PLoS neglected tropical diseases. 2019 Mar 11; 13(3):e0007249. <u>https://doi.org/10.1371/journal.pntd.0007249</u> PMID: <u>30856178</u>
- 13. Biswas A, Wig N, Aggarwal P, Sood R, Wali JP(1999). A new focus of visceral leishmaniasis in sub-Himalayan (Kumaon) region of northern India. J Commun Dis. 1999 Jun; 31(2):73–7. PMID: <u>10810593</u>
- Chappuis F, Rijal S, Soto A, Menten J, Boelaert M (2006). A meta-analysis of the diagnostic performance of the direct agglutination test and rK39 dipstick for visceral leishmaniasis. BMJ. 2006 Oct 7; 333 (7571):723. Epub 2006 Aug 1. <u>https://doi.org/10.1136/bmj.38917.503056.7C</u> PMID: <u>16882683</u>
- Hasker E, Kansal S, Malaviya P, Gidwani K, Picado A, Singh RP, Chourasia A, Singh AK, Shankar R, Menten J, Wilson ME, Boelaert M, Sundar S(2013). Latent infection with Leishmania donovani in highly endemic villages in Bihar, India. PLoSNegl Trop Dis. 2013; 7(2):e2053. <u>https://doi.org/10.1371/journal.pntd.0002053</u> Epub 2013 Feb 14.
- 16. Chakravarty J, Hasker E, Kansal S, Singh OP, Malaviya P, Singh AK, Chourasia A, Singh T, Sudarshan M, Singh AP, Singh B, Singh RP, Ostyn B, Fakiola M, Picado A, Menten J, Blackwell JM, Wilson ME, Sacks D, Boelaert M, Sundar S (2019). Determinants for progression from asymptomatic infection to symptomatic visceral leishmaniasis: A cohort study. PLoSNegl Trop Dis. 2019 Mar 27; 13(3):e0007216. https://doi.org/10.1371/journal.pntd.0007216 eCollection 2019 Mar.
- Hirve S, Boelaert M, Matlashewski G, Mondal D, Arana B, Kroeger A, Olliaro P (2016). Transmission Dynamics of Visceral Leishmaniasis in the Indian Subcontinent—A Systematic Literature Review. PLoSNegl Trop Dis. 2016 Aug 4; 10(8):e0004896. <u>https://doi.org/10.1371/journal.pntd.0004896</u> eCollection 2016 Aug
- Das VN, Siddiqui NA, Verma RB, Topno RK, Singh D, Das S, Ranjan A, Pandey K, Kumar N, Das P (2011). Asymptomatic infection of visceral leishmaniasis in hyperendemic areas of Vaishali district, Bihar, India: a challenge to kala-azar elimination programmes. Trans R Soc Trop Med Hyg. 2011 Nov; 105(11):661–6. Epub 2011 Sep 25. https://doi.org/10.1016/j.trstmh.2011.08.005 PMID: 21945327
- Harith AE, Kolk AH, Kager PA, Leeuwenburg J, Muigai R, Kiugu S, Kiugu S, Laarman JJ (1986). A simple and economical direct agglutination test for serodiagnosis and sero-epidemiological studies of visceral leishmaniasis. Trans R Soc Trop Med Hyg. 1986; 80(4):583–36. <u>https://doi.org/10.1016/0035-9203(86)90149-5</u> PMID: <u>3101241</u>
- Boelaert M, El-Safi S, Hailu A, Mukhtar M, Rijal S, Sundar S, Wasunna M, Aseffa A, Mbui J, Menten J, Desjeux P, Peeling RW(2008). Diagnostic tests for kala-azar: a multi-centre study of the freeze-dried DAT, rK39 strip test and KAtex in East Africa and the Indian subcontinent. Trans R Soc Trop Med Hyg. 2008 Jan; 102(1):32–40. Epub 2007 Oct 22. <u>https://doi.org/10.1016/j.trstmh.2007.09.003</u> PMID: <u>17942129</u>
- Chappuis F, Mueller Y, Nguimfack A, Rwakimari JB, Couffignal S, Boelaert M, Cavailler P, Loutan L, Piola P(2005). Diagnostic accuracy of two rK39 antigen-based dipsticks and the formol gel test for rapid diagnosis of visceral leishmaniasis in northeastern Uganda. J Clin Microbiol. 2005 Dec; 43(12):5973–7. https://doi.org/10.1128/JCM.43.12.5973-5977.2005 PMID: 16333084
- Sivakumar R, Dey A, Sharma P, Singh S(2008). Expression and characterization of a recombinant kinesin antigen from an old Indian strain (DD8) of Leishmania donovani and comparing it with a commercially available antigen from a newly isolated (KE16) strain of L. donovani. Infect Genet Evol. 2008 May; 8(3):313–22. Epub 2008 Feb 15. https://doi.org/10.1016/j.meegid.2008.02.004 PMID: https://doi.org/10.1016/j.meegid.2008 PMID: https://doi.org/10.1016/j.meegid.2008 PMID: https://doi.org/10.1016/j.meegid.2008 PMID: htttps://doi.000 PM
- Sundar S, Singh RK, Bimal SK, Gidwani K, Mishra A, Maurya R, Singh SK, Manandhar KD, Boelaert M, Rai M.(2008) Comparative evaluation of parasitology and serological tests in the diagnosis of visceral leishmaniasis in India: a phase III diagnostic accuracy study. Trop Med Int Health. 2007 Feb; 12 (2):284–9. <u>https://doi.org/10.1111/j.1365-3156.2006.01775.x</u> PMID: <u>17300637</u>
- 24. Ismail A, Gadir AF, Theander TG, Kharazmi A, El Hassan(2006). Pathology of post-kala-azar dermal leishmaniasis: a light microscopical, immunohistochemical, and ultrastructural study of skin lesions and draining lymph nodes. J CutanPathol. 2006 Dec; 33(12):778–87.
- Costa SR, D'Oliveira A Jr, Bacellar O, Carvalho EM (1999). T cell response of asymptomatic Leishmania chagasi infected subjects to recombinant leishmania antigens. Mem Inst Oswaldo Cruz. 1999 May-Jun; 94(3):367–70. <u>https://doi.org/10.1590/s0074-02761999000300015</u> PMID: <u>10348984</u>

- Tsuboi I, Sagawa K, Shichijo S, Yokoyama M M, Ou D W, and Wiederhold M D(1995). Adenosine deaminase isoenzyme levels in patients with human T-cell lymphotropic virus type 1 and human immunodeficiency virus type 1 infections. Clin Diagn Lab Immunol. 1995 Sep; 2(5): 626–630. PMID: 8548545
- Ri G, Ohno S, Furutani M, Furutani Y, Tsukahara T, Hagita N, Haruyama H, Nakamura S, Yamamoto T, Matsuoka R(2010). An indication for correlation between the serum ADA level and gastric cancer risk. Anticancer Res. 2010 Jun; 30(6):2347–9. PMID: 20651391
- Conlon BA, Law WR (2004). Macrophages are a source of extracellular adenosine deaminase-2 during inflammatory responses. Clin Exp Immunol. 2004 Oct; 138(1):14–20. <u>https://doi.org/10.1111/j.1365-2249.2004.02591.x</u> PMID: <u>15373900</u>
- Bimal S, Sinha S, Singh SK, Narayan S, Kumar V, Verma N, Ranjan A, Sinha PK, Das VN, Pandey K, Kar SK, Das P (2012). Leishmania donovani: CD2 biased immune response skews the SAG mediated therapy for a predominant Th1 response in experimental infection. Exp Parasitol. 2012 Jul; 131(3):274– 82. Epub 2012 May 10. https://doi.org/10.1016/j.exppara.2012.04.007 PMID: 22580024
- Murray HW, Delph-Etienne S. Roles of endogenous gamma interferon and macrophage microbicidal mechanisms in host response to chemotherapy in experimental visceral leishmaniasis. Infect Immun. 2000 Jan; 68(1):288–93. <u>https://doi.org/10.1128/iai.68.1.288-293.2000</u> PMID: <u>10603400</u>
- **31.** Murphy ML, Wille U, Villegas EN, Hunter CA, Farrell JP (2001). IL-10 mediates susceptibility to Leishmania donovani infection. Eur J Immunol. 2001 Oct; 31(10):2848–56. <u>https://doi.org/10.1002/1521-4141(2001010)31:10<2848::aid-immu2848>3.0.co;2-t</u> PMID: <u>11592059</u>
- Ansari NA, Saluja S, Salotra P (2006). Elevated levels of interferon-gamma, interleukin-10, and interleukin-6 during active disease in Indian kala azar. Clin Immunol. 2006 Jun; 119(3):339–45. Epub 2006 Mar 15. https://doi.org/10.1016/j.clim.2006.01.017 PMID: 16540374
- Zink RM, Dittmann DL. GENE FLOW, REFUGIA, AND EVOLUTION OF GEOGRAPHIC VARIATION IN THE SONG SPARROW (MELOSPIZA MELODIA). Evolution. 1993 Jun; 47(3):717–729. <u>https://doi.org/10.1111/j.1558-5646.1993.tb01228.x</u> PMID: 28567898
- 34. Ribeiro-de-Jesus A, Almeida RP, Lessa H, Bacellar O, Carvalho EM (1998). Cytokine profile and pathology in human leishmaniasis. Braz J Med Biol Res. 1998 Jan; 31(1):143–8. <u>https://doi.org/10.1590/s0100-879x1998000100020</u> PMID: 9686192
- SINGH Ashish K., et al. Cytokines and chemokines differentially regulate innate immune cell trafficking during post kala-azar dermal leishmaniasis. Journal of cellular biochemistry, 2018, 119.9: 7406–7418.
- 36. Verma S, Kumar R, Katara GK et al. Quantification of parasite load in clinical samples of leishmaniasis patients: II-10 level correlates with parasite load in visceral leishmaniasis. PLOS ONE 2010; 5:e10107 <u>https://doi.org/10.1371/journal.pone.0010107</u> PMID: 20404924
- 37. Kaushal H., Bras-Gonc alves R., Avishek K., Kumar Deep D., Petitdidier E., Lemesre J.-L., Papierok G., Kumar S., Ramesh V and Salotra P. Evaluation of cellular immunological responses in mono- and polymorphic clinical forms of post-kala-azar dermal leishmaniasis in India. 2016 British Society for Immunology, Clinical and Experimental Immunology, 185: 50–60
- Vidya Nand Rabi Das, Niyamat Ali Siddiqui, BiplabPal, Chandra Shekhar Lal, NeenaVerma, AshishKumar, Rakesh Bihari Verma, DhirendraKumar, PradeepDas, Krishna Pandey. To evaluate efficacy and safety of amphotericin B in two different doses in the treatment of post kala-azar dermal leishmaniasis (PKDL). 2017. https://doi.org/10.1371/journal.pone.0174497
- Moulik Srija, Chaudhuri Surya Jyati, Sardar Bikash, Ghosh Manab, Saha Bibhuti, Das NilayKanti, and Chatterjee Mitali. Monitoring of Parasite Kinetics in Indian Post–Kala-azar Dermal Leishmaniasis. • CID 2018: 66
- 40. Costa-Pereira Christiane, Moreira Marcela L, Soares Rodrigo P, Marteleto Bruno H, Ribeiro Vitor M, França-Dias Michelle H, Cardoso Ludmila M, Viana Kelvinson F, Giunchetti Rodolfo C, Martins-Filho Olindo A, SAraújo Márcio S (2015). One-year timeline kinetics of cytokine-mediated cellular immunity in dogs vaccinated against visceral leishmaniasis, December 2015, 11:92
- Murray HW, Flanders KC, Donaldson DD, Sypek JP, Gotwals PJ, Liu J, Ma X (2005). Antagonizing deactivating cytokines to enhance host defense and chemotherapy in experimental visceral leishmaniasis. Infect Immun. 2005 Jul; 73(7):3903–11. <u>https://doi.org/10.1128/IAI.73.7.3903-3911.2005</u> PMID: <u>15972476</u>
- Sypek JP, Chung CL, Mayor SE, Subramanyam JM, Goldman SJ, Sieburth DS, Wolf SF, Schaub RG (1993). Resolution of cutaneous leishmaniasis: interleukin 12 initiates a protective T helper type 1 immune response. J Exp Med. 1993 Jun 1; 177(6):1797–802. <u>https://doi.org/10.1084/jem.177.6.1797</u> PMID: 8098733
- 43. Taylor AP, Murray HW (1997). Intracellular antimicrobial activity in the absence of interferon-gamma: effect of interleukin-12 in experimental visceral leishmaniasis in interferon-gamma gene-disrupted mice. J Exp Med. 1997 Apr 7; 185(7):1231–9. <u>https://doi.org/10.1084/jem.185.7.1231</u> PMID: <u>9104810</u>

- Alvar J, Vélez ID, Bern C, Herrero M, Desjeux P, Cano J, Jannin J, den Boer M; WHO Leishmaniasis Control Team (2018). Leishmaniasis worldwide and global estimates of its incidence. PLoS One. 2012; 7(5):e35671. Epub 2012 May 31. <u>https://doi.org/10.1371/journal.pone.0035671</u> PMID: <u>22693548</u>
- Sundar S, Chatterjee M (2006). Visceral leishmaniasis—current therapeutic modalities. Indian J Med Res. 2006 Mar; 123(3):345–52. PMID: <u>16778315</u>
- 46. Thakur CP, Narayan S, Ranjan A (2004). Epidemiological, clinical & pharmacological study of antimony-resistant visceral leishmaniasis in Bihar, India. Indian J Med Res. 2004 Sep; 120(3):166–72. PMID: <u>15489554</u>
- 47. Amit A, Vijayamahantesh, Dikhit MR, Singh AK, Kumar V, Suman SS, Singh A, Kumar A, Thakur AK, Das VR, Das P, Bimal S (2017). Immunization with Leishmania donovani protein disulfide isomerase DNA construct induces Th1 and Th17 dependent immune response and protection against experimental visceral leishmaniasis in Balb/c mice. Mol Immunol. 2017 Feb; 82:104–113. Epub 2017 Jan 5. https://doi.org/10.1016/j.molimm.2016.12.022 PMID: 28064069
- Vijayamahantesh, Amit A, Kumar S, Dikhit MR, Jha PK, Singh AK, Sinha KK, Pandey K, Das VN, Das P, Bimal S (2015). Up regulation of A2B adenosine receptor on monocytes are crucially required for immune pathogenicity in Indian patients exposed to Leishmania donovani. Cytokine. 2016 Mar; 79:38–44. Epub 2015 Dec 31. <u>https://doi.org/10.1016/j.cyto.2015.12.016</u> PMID: <u>26748211</u>
- 49. Gakis C, Cappio-Borlino A, Pulina G (1998). Enzymes (isoenzyme system) as homeostatic mechanisms the isoenzyme (ADA2) of adenosine deaminase of human monocytes-macrophages as a regulator of the 2'deoxyadenosine. Biochem Mol Biol Int. 1998 Oct; 46(3):487–94. <u>https://doi.org/10.1080/15216549800204012</u> PMID: 9818088
- 50. Zijlstra EE, el-Hassan AM (2001). Leishmaniasis in Sudan. Post kala-azar dermal leishmaniasis. Trans R Soc Trop Med Hyg. 2001 Apr; 95 Suppl1:S59–76
- Sundar S, Rai M (2002). Laboratory diagnosis of visceral leishmaniasis. Clin Diagn Lab Immunol. 2002 Sep; 9(5):951–8. https://doi.org/10.1128/cdli.9.5.951-958.2002 PMID: <u>12204943</u>
- Allain DS, Kagan IG (1975). A direct agglutination test for leishmaniasis. Am J Trop Med Hyg. 1975 Mar; 24(2):232–6. https://doi.org/10.4269/ajtmh.1975.24.232 PMID: 804268
- 53. el Safi SH, Evans DA (1989). A comparison of the direct agglutination test and enzyme-linked immunosorbent assay in the sero-diagnosis of leishmaniasis in the Sudan. Trans R Soc Trop Med Hyg. 1989 May-Jun; 83(3):334–7. <u>https://doi.org/10.1016/0035-9203(89)90493-8</u> PMID: <u>2617579</u>
- 54. el Harith A, Slappendel RJ, Reiter I, van Knapen F, de Korte P, Huigen E, Kolk AH (1989). Application of a direct agglutination test for detection of specific anti-Leishmania antibodies in the canine reservoir. J Clin Microbiol. 1989 Oct; 27(10):2252–7. PMID: <u>2685025</u>
- el Harith A, Kolk A H, Leeuwenburg J, Muigai R, Huigen E, Jelsma T, and Kager P A, Improvement of a direct agglutination test for field studies of visceral leishmaniasis. J Clin Microbiol. 1988 Jul; 26(7): 1321–1325 PMID: 3410946
- Singh A. K., Das V. N. R., Amit A., Dikhit M. R., Mahantesh V., Kumar A., et al. (2018). Identification of clinical immunological determinants in asymptomatic VL and post kala-azar dermal leishmaniasis patients. Iran J. Parasitol. 13, 541–548. PMID: <u>30697307</u>
- 57. Sudarshan M, Singh T, Singh AK, Chourasia A, Singh B, Wilson ME, Chakravarty J, Sundar S (2014). Quantitative PCR in epidemiology for early detection of visceral leishmaniasis cases in India. PLoSNegl Trop Dis. 2014 Dec 11; 8(12):e3366. <u>https://doi.org/10.1371/journal.pntd.0003366</u> eCollection 2014 Dec.
- Sudarshan M, Sundar S (2014). Parasite load estimation by qPCR differentiates between asymptomatic and symptomatic infection in Indian visceral leishmaniasis. Diagn Microbiol Infect Dis. 2014 Sep; 80 (1):40–2. Epub 2014 Mar 7. <u>https://doi.org/10.1016/j.diagmicrobio.2014.01.031</u> PMID: <u>25023070</u>
- Kurtzhals JA, Hey AS, Theander TG, Odera E, Christensen CB, Githure JI, Koech DK, Schaefer KU, Handman E, Kharazmi A (1992). Cellular and humoral immune responses in a population from the Baringo District, Kenya to Leishmania promastigote lipophosphoglycan. Am J Trop Med Hyg. 1992 Apr; 46 (4):480–8. <u>https://doi.org/10.4269/ajtmh.1992.46.480</u> PMID: <u>1575296</u>
- Thammavongsa V, Kern JW, Missiakas DM, Schneewind O (2009). Staphylococcus aureus synthesizes adenosine to escape host immune responses. J Exp Med. 2009 Oct 26; 206(11):2417–27. Epub 2009 Sep 28. <u>https://doi.org/10.1084/jem.20090097</u> PMID: <u>19808256</u>
- Mahamed DA, Mills JH, Egan CE, Denkers EY, Bynoe MS (2012). CD73-generated adenosine facilitates Toxoplasma gondii differentiation to long-lived tissue cysts in the central nervous system. Proc Natl Acad Sci U S A. 2012 Oct 2; 109 (40):16312–7. Epub 2012 Sep 17. <u>https://doi.org/10.1073/pnas.</u> 1205589109 PMID: 22988118
- 62. Vijayamahantesh, Amit A, Dikhit MR, Singh AK, Venkateshwaran T, Das VNR, Das P, Bimal S (2017). Immuno-informatics based approaches to identify CD8+ T cell epitopes within the Leishmania donovani

3-ectonucleotidase in cured visceral leishmaniasis subjects. Microbes Infect. 2017 Jun; 19(6):358–369. Epub 2017 Apr 1. https://doi.org/10.1016/j.micinf.2017.03.002 PMID: 28373107

- 63. Zuckerman SH, Douglas SD. Adenosine kinase activity during in vitro culture of human peripheral blood monocytes and pulmonary alveolar macrophages. Cellular immunology. 1980 Dec 1; 56(2):478–86. <u>https://doi.org/10.1016/0008-8749(80)90122-7</u> PMID: <u>6258808</u>
- 64. Chalmin F, Mignot G, Bruchard M, Chevriaux A, Vegran F, Hichami A, Ladoire S, Derangere V, Vincent J, Masson D, Robson SC, Eberl G, Pallandre JR, Borg C, Ryffel B, Apetoh L, Rebe C, Ghiringhelli F (2012). Stat3 and Gfi-1 transcription factors control Th17 cell immunosuppressive activity via the regulation of ectonucleotidase expression. Immunity 2012 Mar23; 36(3):362–73. Epub 2012 Mar 8. https://doi.org/10.1016/j.immuni.2011.12.019 PMID: 22406269
- Niemand C, Nimmesgern A, Haan S, Fischer P, Schaper F, Rossaint R, Heinrich PC, Müller-Newen G. (2003) Activation of STAT3 by IL-6 and IL-10 in primary human macrophages is differentially modulated by suppressor of cytokine signaling 3. J Immunol. 2003 Mar 15; 170(6):3263–72. <u>https://doi.org/10.4049/jimmunol.170.6.3263</u> PMID: <u>12626585</u>
- 66. Chakravarty J, Hasker E, Kansal S, Singh OP, Malaviya P, Singh AK, Chourasia A, Singh T, Sudarshan M, Singh AP, Singh B. Determinants for progression from asymptomatic infection to symptomatic visceral leishmaniasis: A cohort study. PLoS neglected tropical diseases. 2019 Mar 27; 13(3):e0007216. https://doi.org/10.1371/journal.pntd.0007216 PMID: 30917114